Author's response to reviews

Title: Ecthyma gangrenosum without bacteremia in a previously healthy man: a case report

Authors:

Serap Gencer MD (segencer@tnn.net)
Serdar Ozer MD (ozerserdar@yahoo.com)
Aylin Egegul MD (segencer@tnn.net)
Mustafa Dogan MD (drmustafa395@mynet.com)
Oznur Ak MD (akoznur@yahoo.com)

Version: 5 Date: 12 January 2008

Author's response to reviews:

Dear Editor,

I made the changes that you wanted.

I removed the authors' qualifications (e.g. PhD / MD) from the submissions system.

I gave a different email address for me and added email addresses of the authors on the title page, beneath the corresponding author details.

I placed "Competing interests" before the 'Authors' contributions'.

I didn't placed an Acknowledgements section at the end of the manuscript text because there was no one who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content or who contributed materials essential for the study.

Sincerely,

Dr Serap Gencer